**Author's response to reviews**

**Title:** Systematic Development of the YouRAction program, a computer-tailored Physical Activity promotion intervention for Dutch adolescents, targeting personal motivations and environmental opportunities.

**Authors:**

Richard G Prins (r.prins@erasusmc.nl)
Pepijn van Empelen (p.vanempelen@erasusmc.nl)
Mariëlle A Beenackers (m.beenackers@erasusmc.nl)
Johannes Brug (j.brug@vumc.nl)
Anke Oenema (a.oenema@erasusmc.nl)

**Version:** 4  **Date:** 11 August 2010

**Author's response to reviews:**

Dear dr. Melissa Norton,

Herewith we submit the final revisions of our manuscript. We made some typographical changes.

Sincerely yours on behalf of all authors,

R.G. Prins, MSc
Erasmus Medical Center
Department of Public Health
PO box 2040
3000 CA Rotterdam